#### **ORIGINAL RESEARCH**



# Economic Policy Uncertainty and Volatility Spillovers Among International Stock Market Indices During the COVID-19 Outbreak

Fei Su<sup>1</sup> · Feifan Wang<sup>2</sup> · Yahua Xu<sup>3</sup>

Accepted: 24 February 2024 © The Author(s), under exclusive licence to Springer Japan KK, part of Springer Nature 2024

### Abstract

Using a sample of 16 international stock market indices spanning the period of January 2015 to June 2022, we examine how global equity markets interact with respect to volatility spillover, with a special focus on types of investment horizons, and how the connectedness structure evolves during the COVID-19 outbreak. Empirical results suggest that there is strong evidence of volatility spillovers among global stock markets, and the COVID-19 pandemic further strengthens such volatility spillovers. However, the structure of the frequency connectedness changes gradually when compared to the full sample period. We further investigate if economic policy uncertainty (EPU) affects volatility spillovers among global stock markets. The results suggest that EPU significantly affects the connectedness among global stock markets, particularly during the COVID-19 pandemic period. Overall, the findings suggest that volatility spillovers across international stock markets vary with time horizons and market conditions, which contributes to the academic literature on modelling global volatility spillovers. Practically, the findings of the study contribute to investors and policymakers in adjusting trading strategies and monitoring market risks.

**Keywords** Global stock market · Volatility spillovers · Economic policy uncertainty · COVID-19 pandemic

JEL Classification  $\ C50 \cdot C53 \cdot G14 \cdot G15$ 

Fei Su fei.su@nuaa.edu.cn

<sup>&</sup>lt;sup>1</sup> College of Economics and Management, Nanjing University of Aeronautics and Astronautics, Nanjing 211106, China

<sup>&</sup>lt;sup>2</sup> School of Finance, Shanghai University of Finance and Economics, Shanghai 200433, China

<sup>&</sup>lt;sup>3</sup> China Economics and Management Academy, School of Innovation and Development, Central University of Finance and Economics, Beijing 100081, China

### 1 Introduction

The concept of connectedness of financial markets is considered to be the core of modern finance, particularly in financial risk measurement and management (Diebold & Yilmaz, 2012, 2014). The increasingly deepened integration of financial markets has enhanced the cross-market and cross-border information flows in complex financial systems. Higher financial market integration promotes further development and maturity of financial markets in different countries (regions), but at the same time magnifies the contagion effects of financial risks. Besides, the presence of heterogeneous market participants (e.g., long-term investors versus short-term speculators and traders) would lead to a dissimilarity in return and volatility connectedness between the short-term and long-term horizons. Recently, the financial market's reaction to the COVID-19 global pandemic outbreak since February 2020 has further led to a structural change in the connectedness among financial assets and across financial markets.

It's worth mentioning that the literature on the economic impacts of the COVID-19 has been burgeoning since 2020.<sup>1</sup> The outbreak of COVID-19, which has been perceived as a "black swan" event, has attracted much attention from researchers and posed numerous new questions regarding financial contagion in the era of COVID-19 (Yarovaya et al., 2021a, 2021b, 2022). The relevant research has widely examined the financial impacts of the COVID-19 pandemic and suggested that the COVID-19 outbreak induced finance contagion by affecting risk tolerance and uncertainty among investors in the crisis period (Akhtaruzzaman et al., 2022; Corbet et al., 2022; Guo et al., 2021; Iyer & Simkins, 2022; Nguyen et al., 2021; Su et al., 2023; Yarovaya et al., 2021a, 2021b). For example, Sharif et al. (2020) argue that COVID-19 outbreak has a greater effect on the US geopolitical risk and economic uncertainty than on the US stock market itself. The extant literature has also widely examined the impacts of economic policy uncertainty (EPU) on risk contagion in financial markets (Albulescu et al., 2019; Karanasos & Yfanti, 2020; Wang et al., 2019; Zhang & Yan, 2020; etc.). For example, Qian et al. (2020) find that most of the abnormal market correlations exist in periods with high levels of economic policy uncertainties (EPU). Similarly, Dai et al. (2021) show that the crash risk of the US stock market is more affected by EPU during the COVID-19 pandemic. However, there are few papers that investigate EPU's role in driving the dynamic volatility spillovers among global stock markets before and during the COVID-19 pandemic crisis.

Under this background, in this study we examine the volatility connectedness (a.k.a., volatility spillovers) of leading global stock market indices over a full sample period and a sub-period corresponding to the COVID-19 outbreak.<sup>2</sup>

<sup>&</sup>lt;sup>1</sup> Several special issues on the economic and financial impacts of COVID-19 have been arranged and appeared in a large number of reputable economics and finance journals (i.e., Finance Research Letters, International Review of Financial Analysis, Journal of Economics and Business, etc.) over the past few years.

<sup>&</sup>lt;sup>2</sup> In this study, we interchangeably use the term "spillover" and "connectedness".

Methodologically, we apply the frequency-based connectedness approach of Baruník & Krehlík (2018), which allows us to uncover volatility spillovers while accounting for the investment horizons of heterogeneous market participants that often vary across frequencies. In addition, we explore the impact of economic policy uncertainty (EPU) on connectedness among global stock markets by utilizing time series regression. We particularly focus on how EPU affects the interactions between global stock markets (i.e., volatility connectedness) during the COVID-19 pandemic.

Our study offers several contributions. First, we extend the existing literature arguing that volatility spillovers across international stock markets vary with time horizons (e.g., Cheng et al., 2022; Ciaian et al., 2018; Ji et al., 2020), by using the novel methodology of Baruník and Krehlík (2018) which distinguishes between high-, medium-, and low-frequency bands that can reflect different investment horizons. This is important as value investors have long-term investment horizons, whereas speculators and high-frequency traders have short-term trading horizons that are reflected in high frequencies (Bouri et al., 2020). Second, we provide a discussion of the event of COVID-19 global pandemic and how it translates into the dynamics of volatility connectedness among global stock markets. Overall, the structural changes in return and volatility connectedness, as reactions to the COVID-19 pandemic, provides an ideal setting for examining the impact of COVID-19 pandemic on financial markets. Third and finally, we investigate EPU's role in driving the dynamic volatility spillover between global stock markets before and during the COVID-19 pandemic crisis, which have important implications for retail investors, portfolio managers, policymakers, and central banks.

The remainder of this paper proceeds as follows. Section 2 presents a brief review on extant studies of volatility connectedness and the impact of COVID-19 pandemic on volatility spillovers. Section 3 describes the data and introduces Baruník and Krehlík (2018)'s frequency connectedness methodology. Section 4 presents our main empirical results and robustness checks. Section 5 concludes with policy implications.

# 2 Literature Review

A vast amount of researchers have continuously proposed extended models of the Diebold and Yilmaz (2009, 2012, 2014) methodology to construct return and volatility spillover indices. Most of those studies use autoregressive (AR) model, vector autoregressive (VAR) model, multivariate GARCH-type models, or copula models (e.g., Aslanidis et al., 2019; Bouri et al., 2020; Ciaian et al., 2018; Ji et al., 2019a, 2019b, 2020; Katsiampa, et al., 2019; Moratis, 2021; Tiwari et al., 2020). However, those methods proposed are not suitable to study volatility spillovers and connectedness in a time-varying setting which differentiates between various investment horizons from short- to long-term. Baruník et al. (2016, 2017) propose the method of asymmetric volatility spillover by combining the Diebold and Yilmaz (2009)' connectedness measurement and realized semi-variance (RSV) which differentiates the variance caused by positive and negative returns, and can quantify asymmetric connectedness more flexibly. Moreover, Baruník and Krehlík (2018) propose a new framework of variance decomposition spectrum representation, which can split the measurement of financial market connectedness into several cycles or different frequency bands. So far, discussions about the volatility spillovers among global stock markets under various investment horizons have been very limited.

With the spread of the COVID-19 pandemic and its increasing impacts on global equity markets, Corbet et al. (2022) investigate the existence of contagion between Chinese stock markets during the COVID-19 pandemic and find that COVID-19 has a significantly positive impact on the volatility of the Shanghai and Shenzhen Stock Exchanges. Rehan et al. (2022) examine the short-term stress of COVID-19 pandemic on 32 major stock markets covering American, European, Asian and Pacific countries. Das et al. (2023) investigate the downside potential of G7 stock markets in face of COVID-19 relative to other economic crises. Shen and Zhang (2021) explore the distinct market reactions to the COVID-19 outbreak in the Chinese stock market. Their results reveal that the go-outsides (GO) stocks suffer significant losses on the event date, while for the stay-at-home (SAH) stocks, no significantly negative return is observed on the event date. In the same vein, Ashraf (2020) document that stock markets respond negatively to the increases in daily confirmed COVID-19 cases.

The extant literature has also examined volatility spillovers among stock markets during the COVID-19 outbreak, including global stock markets (BenSaida et al., 2018; Gamba-Santamaria et al., 2019), Asia-Pacific stock markets (Li et al., 2023), U.S. and BRICS stock markets (Mclver & Kang, 2020). Yadav et al. (2023) apply Granger causality and dynamic conditional correlation generalized autoregressive conditional heteroscedasticity (DCC-GARCH) to investigate the spillover between Chinese and emerging economies. The result reveals that there is no spillover effect from China to emerging economies in full period (i.e., from January 3, 2000 to February 7, 2020), expect for Indonesian stock market. Liu et al. (2021) and Lin and Chen (2021) utilize econometric models to evaluate the impact of the COVID-19 outbreak on major global stock market indices. Their empirical results show that all major global stock market indices fell sharply immediately after the virus outbreak. Zhang et al., (2020a, 2020b) investigate the general pattern of country-specific risk and systematic risk in the global financial markets during the COVID-19 pandemic and find evidence of increasing financial market volatility due to the pandemic. So et al. (2021) examine the impact of the COVID-19 pandemic on Hong Kong stock market connectedness and imply increased network connectedness in financial networks during the COVID-19 outbreak.

During financial crisis and market stress periods, the degree of spillovers in financial markets could also be influenced by economic policy uncertainty (EPU). In this context, previous studies have documented that the EPU index as developed by Baker et al. (2016) affects stock market returns (Antonakakis et al., 2013; Arouri et al., 2016; Christou et al., 2017; Guo et al., 2021; He et al., 2020; Hu et al., 2018; Phan et al., 2018; Xiong et al., 2018) and volatility (Balcilar et al., 2019; Mei et al., 2018; Yu & Song, 2018; Yu et al., 2022; Wang et al., 2020). A recent strand of literature has focused on the impact of EPU on the co-movement among various financial assets, including connectedness among regional stock markets (Li et al., 2015; Li & Peng, 2017), bond and stock markets (Fang et al., 2017), commodity and stock markets (Badshah et al., 2019), and Bitcoin and conventional financial markets (Fang et al., 2018; Matkovskyy et al., 2020). Most of these studies find evidence of a significantly negative relationship between EPU and the co-movement between these variables, which highlights a significant portfolio implication related to EPU (Badshah et al., 2019). Li et al. (2021) unveil the impact of EPU on the financial markets in the Asia–Pacific region and find that the correlations between EPU and financial network connections have risen after the COVID-19 outbreak.

Another strand of literature investigates the impact of EPU on volatility spillovers. For example, the relevant studies have been focusing on the impact of EPU among global stock markets (Shi & Wang, 2023), the spillover effect of U.S. EPU in the post-epidemic era (Li et al., 2023), the asymmetric volatility spillovers between international EPU and the U.S. stock market (He et al., 2020), and the driving force of EPU on volatility spillovers in electricity markets (Ma et al., 2022). However, there are few papers that investigate EPU's role in driving the dynamic volatility spillovers among global stock markets before and during the COVID-19 pandemic crisis.

The above literature review points to three research gaps which we attempt to address in this study. First, we extend the existing literature by analyzing timevarying volatility spillovers across international stock markets, based on the novel methodology of Baruník and Krehlík (2018) which distinguishes between high, medium, and low frequency bands that reflect different types of investment horizons, which is crucial to investment decisions regarding portfolio diversification and hedging strategies. Second, there is a lack of evidence on volatility spillovers among major stock markets during the catastrophic event of the COVID-19 outbreak. Third, whether EPU affects stock market volatility spillovers, particularly in the context of the COVID-19 pandemic, has not been fully explored. Considering the fact that EPU could drive the connectedness among financial variables, it is highly possible that volatility connectedness arises due to heterogeneous frequency responses to shocks in financial variables. Compared to previous studies, our research contributes to the strand of literature about volatility spillovers among major stock markets, by systematically analyzing a comprehensive data set of 16 major stock markets before and during the COVID-19 outbreak, both in the time domain and the frequency domain.

# 3 Data and Methodology

#### 3.1 Data

Our study covers 16 most popular stock market indices worldwide, including China, India, and 14 developed markets, spanning the period from 2015 to 2022. For a full list of the stock market indices used in this study, please see Table 6 in the Appendix.

The daily price data is downloaded from the Oxford-Man Institute's realized library (https://realized.oxford-man.ox.ac.uk/data). We compute daily logarithmic returns and realized variances of major global stock market indices based on intraday data.<sup>3</sup> A list of stock market indices utilized in this study is provided in Table 6 in the Appendix. To examine the impact of economic and policy uncertainty on global stock market connectedness, the global economic policy uncertainty index (GEPU) is retrieved from Federal Reserve Bank of St. Louis (FRED). We also obtain data on the number of confirmed COVID-19 cases (deaths) per million inhabitants from Johns Hopkins University (https://coronavirus.jhu.edu/data). Finally, we control for the uncertainty index represented by the CBOE market volatility index (VIX). The time series of VIX index is obtained from the website of the Chicago Board Options Exchange (https://www.cboe.com/indices/).

# 3.2 Methodology

### 3.2.1 Measure of Realized Volatility

Following the pioneering work of Andersen and Bolleslev (1998), Andersen (2000), Andersen et al., (2003, 2007) introduce the concept of realized volatility which utilizes the intraday data to construct the model-free estimate of asset price volatility. The realized variance (RV) is calculated as the sum of squared log-price differences sampled over intraday intervals. Following Liu et al. (2015) and Bollerslev et al. (2018), the tick-level data can be aggregated into 5-min intervals which strikes a good trade-off between speed and accuracy.

Note that the realized volatility is approximately free of measurement error only under particular conditions. Therefore, the existing studies have proposed alternative approaches to reduce the impact of the noise term induced by microstructure noise on the estimation of the integrated variance (see, e.g., Ait-Sahalia, et al., 2005; Barndorff-Nielsen & Shephard, 2004, 2006; Barndorff-Nielsen, et al., 2008, 2010). Among the alternative measures, the realized bi-power variation (BV) and realized semivariance (RSV) are the most popular ones, both of which prove to be robust to market microstructure noise. Please refer to McAleer and Medeiros (2008) and Bollerslev et al. (2018) for some useful reviews on measures of realized volatility.

#### 3.2.2 Measure of Volatility Spillover

The traditional understanding of connectedness in financial systems through simple correlations and related measures turns out to be insufficient to explain the

<sup>&</sup>lt;sup>3</sup> In this paper, we adopt three measures appropriate to describe stock return variances, including realized variance, realized semi-variance, and bi-power variation to account for microstructure noise and jumps in stock price process, with the availability of tick-level data. Liu et al. (2015) show that the choice of 5-min sampling frequency can largely circumvent the impact of market microstructure noise. Therefore, all the realized volatility measures are calculated by rolling sampling of transaction price series at 5-min frequency (i.e., 5-min sub-sampled).

systematic dynamics of the complicated networks that modern financial markets own. Following Diebold and Yilmaz (2012, 2014) and Baruník and Krehlík (2018), we examine the volatility connectedness of major international stock market indices through the generalized vector autoregression (VAR) framework and imply forecast error variance decomposition to form a numerical spillover score among a system of variables (Belke et al., 2016; Kumar et al., 2022). Namely, Diebold and Yilmaz (2012) derive the procedure from the standard VAR model as following:

$$x_t = \Phi_1 x_{t-1} + \Phi_2 x_{t-2} + \dots + \Phi_p x_{t-p} + \varepsilon_t$$

where  $\Phi_1,...,\Phi_p$  are coefficient matrices and  $\varepsilon_t$  is the residual term. Assuming that the roots of  $|\Phi_1|$  are outside the unit-circle, the VAR process is represented through the MA( $\infty$ ) process as  $x_t = \Psi(L)\varepsilon_t$ , where  $\Psi(L)$  is a matrix of infinite lag polynomials.

Building on the infinite moving average representation of the VAR model, the generalized forecast error variance decomposition (GFEVD) can be written in the following form:

$$\left(\Phi_{H}\right)_{j,k} = \frac{\sigma_{kk}^{-1} \sum_{h=0}^{H} \left(\left(\Psi_{h} \sum\right)_{j,k}\right)^{2}}{\sum_{h=0}^{H} \left(\left(\Psi_{h} \sum \Psi_{h}'\right)_{i,j}\right)}$$

where  $\Psi_h$  is a (N×N) matrix of moving average coefficients at lag h,  $\sigma_{k,k} = (\Phi_H)_{j,k}$  is the contribution of the  $k^{th}$  variable for the forecast error variance of the element j for H horizon, and  $\Sigma$  is the variance matrix of residuals.<sup>4</sup>

According to Diebold and Yilmaz (2012), we can thus define connectedness measure of the spillover index as the ratio of the sum of the off-diagonal elements to the sum of the whole matrix, which is given by:

$$C_H = 100 \times \frac{\sum_{j \neq k} (\Phi_H)_{j,k}}{\sum \Phi_H} = 100 \times \left(1 - \frac{\operatorname{Tr}\{\Phi_H\}}{\sum \Phi_H}\right)$$

where  $Tr\{\cdot\}$  denotes the trace operator, and the total of all elements of  $\Phi_H$  matrix is presented in the denominator.

In addition, Baruník and Krehlík (2018) use a frequency response function  $\Psi(e^{-i\omega}) = \sum_{h} e^{-i\omega h} \Psi_{h}$  that is obtained from the Fourier transform of the coefficients  $\Psi$ , with *i* being an imaginary constant. Therefore, the frequency-based connectedness for frequency band *d* can be calculated as<sup>5</sup>:

<sup>&</sup>lt;sup>4</sup> For highly correlated financial variables such as stock market returns and volatilities we examine in this study, the variance–covariance matrix  $\Sigma$  is not necessarily a diagonal matrix, so that the shocks in  $\epsilon_t$  are allowed to be contemporaneously correlated. As the values of the entries in each row of the  $\Phi_H$  matrix may not add up to one, each entry is then normalized by scaling the sum of the entries in each row.

<sup>&</sup>lt;sup>5</sup> For more details about the frequency-domain modelling approach, please refer to Baruník and Krehlík (2018).

$$C_d^F = 100 \times \left(\frac{\sum \overline{\theta}_d}{\sum \theta_{\infty}} - \frac{\operatorname{Tr}\{\overline{\theta}_d\}}{\sum \theta_{\infty}}\right) = C_d^w \times \frac{\sum \theta_d}{\sum \theta_{\infty}}$$

where  $Tr\{\cdot\}$  is the trace operator, and the sum of all the elements of  $\Phi_d$  matrix is presented in the denominator.  $C_d^w$  stands for the within connectedness measure, which reflects the connectedness within the specific frequency band. Thus, the frequency-based connectedness splits the overall connectedness into various components (or cycles) that add up to the original connectedness measure.

By adopting the methodology of Baruník and Krehlík (2018) which distinguishes between high, medium, and low frequency bands that can reflect various investment horizons and risk preferences of heterogeneous investors, we can conduct a thorough inspection of the time evolution of dynamic overall and horizon-specific connectedness among major stock markets.

#### 3.2.3 EPU and Dynamic Connectedness Between Stock Markets

Several recent studies have argued that the dependence and connectedness among financial assets and across countries (regions) are affected by EPU (Badshah et al., 2019; Fang et al., 2017, 2018; Li & Peng, 2017; Li et al., 2018; Matkovskyy et al., 2020). These studies document that EPU negatively impacts the correlations between these variables. Therefore, we argue that EPU is a potential common factor that drives the connectedness among financial variables attributed to heterogeneous frequency responses to shocks.

After computing the time-varying spillover indices based on the TVP-VAR model, we examine whether EPU and confirmed COVID-19 cases simultaneously drive the volatility connectedness among global stock markets during the COVID-19 crisis. To this end, we estimate the following model using OLS, which is similar to those of Erdem (2020) and Engelhardt et al. (2021):

 $Connectedness_t = \alpha_t + \beta_1 GEPU_t + \beta_2 Cases_t + \beta_3 GEPU_t * Cases_t + \beta_4 VIX_t + \varepsilon_t$ 

where connectedness is the weekly measure for volatility connectedness among the 16 global stock market indices, calculated as the 5-day moving average of total volatility spillover index with realized variance (RV).<sup>6</sup> Changes in governments' economic policies in different countries are controlled by including the GEPU constructed by Baker et al. (2016), and *Cases<sub>t</sub>* is the logarithm of one plus the number of confirmed COVID-19 cases (deaths) per million inhabitants in week t.<sup>7</sup> VIX is the CBOE stock market volatility index. We also add the interaction terms between EPU

<sup>&</sup>lt;sup>6</sup> In the robustness check, we also use weekly measure for volatility connectedness index with realized variance (RV), realized semi-variance (RSV) and bi-power variation (BV) in time-frequency domain.

<sup>&</sup>lt;sup>7</sup> GEPU is a GDP-weighted average of national EPU indices for 20 countries: Australia, Brazil, Canada, Chile, China, France, Germany, Greece, India, Ireland, Italy, Japan, Mexico, the Netherlands, Russia, South Korea, Spain, Sweden, the United Kingdom, and the United States (Baker et al., 2016, 2022). The EPU index is provided at monthly frequency, while our spillover indices are calculated weekly. Therefore, we interpolate the monthly EPU index to weekly frequency by cubic spline.

and confirmed COVID-19 cases (deaths) per million inhabitants to examine the confounding effects of EPU on volatility connectedness among global stock markets.

# 4 Empirical Results

#### 4.1 Descriptive Statistics

Descriptive statistics of the daily returns and all volatility measures are given in Table 1. All the time series of daily returns are leptokurtic, i.e., with extreme outliers. None of the time series are close to Gaussian distributions, which is confirmed by the statistics of the Jarque–Bera test. Shanghai stock exchange (SSEC) of China has the highest daily average return of 0.13% during our sample period, while NIFTY 50 (NSEI) of India has the lowest daily average return of -0.10%.

Regarding the volatility, we observe that the SSEC of China exhibits the highest average realized volatility, while Madrid general index (SMSI) has the lowest average volatility during our sample period. The magnitude of volatility of Shanghai composite index is almost twice that of Madrid general index. In fact, the investor structure in China's stock market differs substantially from that of developed markets. For example, the ratio of turnover by retail investors in Chinese stock market is much higher than that in developed countries' stock markets. In addition, unlike the US stock market, Chinese investors are prone to buy stocks with recent extreme gains (i.e., chasing winners) (Wang et al., 2022a, 2022b). Therefore, China's stock market is largely dominated by retail investors who prefer to speculate and buy winners, which partially explains the puzzle of excess stock price volatility in China's stock market (Li et al., 2018; Su & Wang, 2021).

However, Russel 2000 (RUT) exhibits the largest maximum value of volatility, indicating frequent volatility jumps. The volatility series are unsurprisingly positively skewed. All the realized volatility series deviate from normal distributions as indicated by the skewness and kurtosis values, as well as by the Jarque–Bera test. The results also show evidence of serial correlations of volatilities as demonstrated by the Ljung-Box test. There is also very strong evidence of stationarity for all the daily time series as indicated by the ADF test.

#### 4.2 Volatility Spillovers for the Full Sample Period

Our full sample covers the period spanning from January 2015 to June 2022. Panel A of Table 2 presents the results of volatility spillover index with realized variance (RV) over the whole sample period by utilizing Diebold and Yilmaz (2012)'s spillover index, which indicates the degree of volatility spillover across all frequency bands. The spillover index lies between 0 and 100. FROM represents shocks from others, TO represents shocks to others, and NET indicates the net directional connectedness of spillover, calculated by subtracting the total directional connectedness "TO" others from "FROM". If NET is greater than zero, it indicates that the country is a net transmitter of shocks during the period, otherwise, it will

	······								
Index	Ν	Mean	Std Dev	Min	Max	Skew	Kurt	ADF	J-B test
Descriptiv	e statis	tics for a	laily return	ı					
AEX	1957	0.14	7.93	-60.38	49.18	-0.78	9.42	-13.14***	0.74***
BFX	1957	-0.02	8.39	- 103.43	46.91	-1.51	19.67	-12.26***	3.23***
FCHI	1957	0.18	8.61	-71.62	57.94	-0.93	11.70	-13.37***	1.15***
FTMIB	1957	-0.33	10.06	- 129.76	37.89	-1.71	19.47	- 12.67***	3.19***
FTSE	1957	0.21	10.42	- 103.00	71.84	-0.90	13.06	-13.34***	1.42***
GSPTSE	1957	-0.21	6.92	-84.53	54.05	-1.81	43.97	-12.83***	15.88***
IXIC	1957	0.23	9.56	-68.31	46.29	-0.83	10.09	-12.86***	0.85***
KS11	1957	-0.60	7.81	- 102.08	54.95	-0.30	9.4	-13.23***	0.72***
SPX	1957	-0.62	10.34	-69.93	51.51	-0.16	5.45	-13.09***	0.24***
N225	1957	-0.11	8.65	-56.02	29.58	-1.47	21.05	-11.84***	3.68***
NSEI	1957	-0.85	9.09	-63.47	95.14	-0.79	8.48	-12.61***	0.61***
OSEAX	1957	0.41	11.10	-98.30	58.31	-1.17	14.18	-12.23***	1.69***
RUT	1957	-0.09	11.59	- 123.79	69.37	-1.49	17.88	- 12.94***	2.68***
SMSI	1957	-0.59	10.07	-90.05	66.92	-0.97	18.11	-12.78***	2.71***
SSEC	1957	1.27	12.18	-70.23	77.51	-1.21	8.04	-11.80***	0.58***
STI	1957	-0.23	6.66	-40.83	48.10	-0.53	9.11	-12.40***	0.69***
Descriptiv	e statis	tics for a	laily realiz	ed variance	e (5-min s	sub-samp	oled)		
AEX	1957	0.77	1.74	0.03	31.83	10.87	153.33	-7.79***	195.96***
BFX	1957	0.75	1.55	0.04	27.09	9.42	118.12	-7.67***	116.90***
FCHI	1957	0.88	1.83	0.05	36.61	10.42	149.44	-7.86***	186.03***
FTMIB	1957	0.89	1.94	0.02	28.13	8.17	88.26	-7.77***	65.83***
FTSE	1957	0.54	0.70	0.04	10.89	5.85	55.48	-6.53***	26.27***
GSPTSE	1957	0.54	0.73	0.05	22.59	16.59	446.51	-7.71***	1638.03***
IXIC	1957	0.61	1.67	0.00	38.14	14.76	264.84	-7.93***	580.24***
KS11	1957	0.80	1.86	0.02	29.53	10.07	126.94	-7.51***	134.97***
SPX	1957	0.72	2.07	0.01	36.91	10.51	137.11	-7.80***	157.22***
N225	1957	0.68	1.99	0.01	47.42	13.61	236.08	-8.04***	461.47***
NSEI	1957	0.51	1.05	0.00	22.78	11.54	181.87	- 8.09***	274.63***
OSEAX	1957	0.49	2.65	0.05	45.31	8.18	91.89	-6.58***	71.18***
RUT	1957	0.65	1.70	0.06	54.90	20.04	562.79	- 8.06***	2601.14***
SMSI	1957	0.42	1.12	0.01	19.89	10.54	136.93	-7.36***	156.83***
SSEC	1957	0.93	1.61	0.06	27.81	8.96	108.22	-8.21***	98.31***
STI	1957	0.82	1.40	0.01	29.79	9.32	136.53	-7.54***	155.15***
Descriptiv	e statis	tics for a	laily realiz	ed semi-va	riance (5	-min sub-	-sampled)		
AEX	1957	0.45	1.11	0.01	23.22	11.59	174.50	-8.32***	253.19***
BFX	1957	0.43	0.95	0.03	19.19	10.59	153.42	-7.54***	196.00***
FCHI	1957	0.50	1.12	0.02	24.31	11.27	175.02	-8.17***	254.47***
FTMIB	1957	0.48	1.35	0.01	46.35	21.98	698.37	-8.30***	4000.93***
FTSE	1957	0.32	0.98	0.01	36.85	27.61	987.28	- 8.67***	7989.30***
GSPTSE	1957	0.35	1.16	0.03	45.72	31.79	1204.74	- 10.66***	11,892.31***
IXIC	1957	0.36	0.98	0.00	27.18	15.69	338.71	- 8.98***	945.43***

Index	Ν	Mean	Std Dev	Min	Max	Skew	Kurt	ADF	J-B test
KS11	1957	0.46	1.05	0.01	21.93	10.37	152.37	-7.71***	193.22***
SPX	1957	0.42	1.37	0.00	28.09	12.92	210.91	-8.28***	368.91***
N225	1957	0.52	2.15	0.01	61.47	19.74	488.44	-8.29***	1962.12***
NSEI	1957	0.28	0.61	0.01	17.61	15.50	366.16	-7.63***	1103.34***
OSEAX	1957	0.44	1.50	0.02	20.89	7.35	71.30	$-6.82^{***}$	43.30***
RUT	1957	0.62	4.50	0.01	187.13	36.88	1506.25	-9.67***	18,582.32***
SMSI	1957	0.25	1.05	0.01	29.91	21.15	536.76	-8.29***	2368.74***
SSEC	1957	0.55	1.97	0.02	59.47	23.75	655.21	-9.60***	3526.24***
STI	1957	0.45	0.81	0.00	17.36	9.93	153.00	-7.47***	194.51***
Descriptiv	e statis	stics for a	laily bi-po	wer vai	riation (5-min	ı sub-sar	npled)		
AEX	1957	0.87	2.10	0.03	42.00	11.97	186.08	- 7.99***	287.61***
BFX	1957	0.85	1.80	0.04	36.89	10.31	146.65	-7.57***	179.20***
FCHI	1957	0.99	2.12	0.04	43.62	10.80	160.48	-7.89***	214.25***
FTMIB	1957	0.94	2.27	0.03	59.63	13.03	263.24	$-8.02^{***}$	571.78***
FTSE	1957	0.62	1.26	0.03	36.96	15.52	380.81	-7.45***	1192.77***
GSPTSE	1957	0.68	1.41	0.06	52.25	26.36	922.00	-9.50***	6968.58***
IXIC	1957	0.70	2.56	0.00	93.36	26.55	896.27	-9.50***	6586.69***
KS11	1957	0.88	1.89	0.03	30.83	9.20	110.24	-7.58***	102.07***
SPX	1957	0.81	2.36	0.01	41.53	11.05	152.64	- 8.06***	194.38***
N225	1957	0.99	2.77	0.01	66.68	14.42	277.47	-7.68***	635.86***
NSEI	1957	0.54	1.08	0.01	26.08	12.54	223.93	-7.48***	414.88***
OSEAX	1957	0.79	2.86	0.05	41.46	7.68	78.68	-6.64***	52.52***
RUT	1957	1.16	5.02	0.07	196.96	31.26	1184.43	-8.20***	11,494.66***
SMSI	1957	0.47	1.49	0.02	36.86	15.44	311.66	-7.77***	801.45***
SSEC	1957	1.03	2.47	0.09	61.05	16.59	362.43	-8.55***	1082.28***
STI	1957	0.88	1.46	0.01	27.28	8.48	105.69	-7.65***	93.64***

This table presents the descriptive statistics of daily returns and realized volatility measures of the 16 major stock market indices. The sample period is January 2015 to Jun 2022. For Mean, Std Dev, Min, and Max, the values are expressed in scientific notation as  $\times 10^{-4}$ . For J-B test, the values are expressed in scientific notation as  $\times 10^{-4}$ . For J-B test, the values are expressed in scientific notation as  $\times 10^{4}$ . \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels respectively. For a full list of the stock market indices adopted in this study, please see Table 6 in the Appendix

be a net receiver of shocks. The results in Panel A of Table 2 show that there is substantial evidence of volatility spillovers in the global stock markets, as the total spillover index is as high as 80.55. In fact, the forecast error variance of each stock market index can be logically divided into two parts—its own contribution and contributions from other stock markets. The untabulated results suggest that the dynamic volatility spillovers in global stock markets are apparently dominated by the own-market structure, as the values in the diagonal are much larger than those in the non-diagonal cells which represents cross-market effects. Among the 16 major stock markets, China's Shanghai composite index (SSEC) exhibits the highest

Table 2	Volatilit	y spillove	sr measur	res for R'	V—DY m	ethod											
	AEX	BFX	FCHI	IXIC	SPX	N225	NSEI	RUT	STI	FTSE	KS11	SSEC	OSEAX	GSPTSE	ISMS	FTMIB	FROM
Panel A:	full sam	ple															
TO	8.11	5.53	7.74	7.69	0.27	2.85	1.17	6.18	10.57	5.04	0.96	0.18	3.20	9.84	2.70	8.53	80.55
FROM	5.52	5.52	5.49	5.27	5.05	4.52	5.73	5.38	5.27	5.30	5.64	2.02	4.61	5.18	4.93	5.13	N.A
NET	2.59	0.01	2.25	2.42	-4.78	- 1.67	-4.56	0.80	5.30	-0.26	-4.68	-1.84	-1.41	4.66	- 2.23	3.40	N.A
Panel B:	COVID	-19 perio	q														
TO	7.08	3.92	6.58	6.32	0.30	5.43	0.35	4.96	11.15	6.17	0.27	0.45	3.85	11.33	7.22	10.66	86.03
FROM	5.67	5.70	5.64	5.56	5.62	4.80	6.09	5.66	5.35	5.42	5.88	3.76	4.80	5.28	5.49	5.32	N.A
NET	1.41	- 1.78	0.94	0.76	-5.32	0.63	- 5.74	-0.70	5.80	0.75	-5.61	- 3.31	-0.95	6.05	1.73	5.34	N.A
This tab Yilmaz ( Tility, we To conse indicates zero, it ii shown in	e presen 2012)'s utilize r rrve spac the net ndicates	ts the res spillover ealized v 2e, the res direction that the c om right	ults of w index in ariance ( sults for ] al conne country is country is	the time the time RV), rea RSV and Ctedness s a net tr s a net tr	spillovers comain. 7 lized semi 1 BV are p of spillov cansmitter t of each p	among glu The COV. -variance rovided ii 'er, calcul of shocks anel	obal stock ID-19 per (RSV), au n Table 7 ated by su ated by su	markets iod in thii ad bi-pow in the Ap in the Ap ibtracting te period,	over the s study is er variat ppendix. The tota otherwis	whole san s defined s ion (BV), FROM re FROM re se, it will	nple perio as spannin all calcula presents s nal conneo be a net r	d and dur ig from Fa ated based hocks fro ctedness ' eceiver of	ing the CC ebruary 4, 1 on rolling m others, 7 TO" other shocks. T	WID-19 per 2020 to June 5 sampling p FO represent 's from "FR he total spil	iod, by ut e 30, 2022 rice series is shocks OM". If N lover inde	2. To meas 2. To meas 8 at 5-min j to others, a VET is gre x is highli	bold and ure vola- ntervals. and NET ater than ghted, as

me	
-DY	
2	
22	
for	
measures	
spillover	
Volatility	
2	

own-market volatility spillovers, followed by Nikkei 225 (N225) and Oslo exchange all-share index (OSEAX). Overall, the cross-market volatility spillover effects are mostly evident for stock markets in the same region. Interestingly, Shanghai Composite Index (SSEC) has the lowest volatility spillover from (to) other major stock markets (i.e., volatility spillover index of 0.18 and 2.02), probably due to strict capital controls in mainland China. For example, Kitano and Zhou (2022) document that China's capital controls are very effective in controlling the volume of equity flows in China, which mitigates the volatility spillover effect.

In addition, Table 7 in the Appendix shows the magnitudes of volatility spillovers among the 16 sample stock markets with realized semi-variance (RSV) and bipower variation (BV) respectively, during the whole sample period and the COVID-19 period. The total spillover index is 76.88 and 82.28 for realized semi-variance and bipower variation respectively, for the full sample period. Unsurprisingly, the connectedness index with bipower variation is substantially higher, which corrects for jumps in stock price process and therefore increases the co-movements between stock returns and volatility. Overall, the results consistently suggest the existence of strong volatility spillovers among global stock markets.

### 4.3 Volatility Spillovers During the COVID-19 Pandemic

In this subsection, we test whether the COVID-19 global pandemic outbreak initially reported since February 2020 has led to a structural change in the volatility connectedness among global financial markets. Specifically, we examine volatility spillovers and their time–frequency dynamics among the 16 major global financial markets from the outbreak of COVID-19 to present. In Panel B of Table 2, the total spillover index measures the average proportion of volatility forecast error variance explained by shocks emanating from other stock markets for the period between February 2020 and June 2022 (i.e., the COVID-19 outbreak period), by utilizing the Diebold and Yilmaz (2012)'s spillover methodology.

Similar to the results as shown in Panel A of Table 2, we observe that own-country stock market spillovers contribute to the largest portion of forecast error variances, as the diagonal cells possess higher values compared to the non-diagonal elements. During the COVID-19 period, the total spillover index with realized variance (RV) increases to approximately 86.03 as shown in Panel B of Table 2. The result indicates that global stock markets are not independent of each other, particularly during turbulent times, and the volatility spillovers accounts for as high as 86% of volatility forecast error variances on average. This large value indicates that the transmission of international stock market shocks is an important source of domestic stock price fluctuations. The result also signifies an intensification of the volatility connectedness during turbulent periods, which is in line with previous studies of fiat currencies (e.g., Greenwood-Nimmo et al., 2016), cryptocurrencies (e.g., Yi et al., 2018; Kumar et al., 2022), and equity markets (Wang et al., 2022a, 2022b; Youssef et al., 2021). During the COVID-19 period, China's Shanghai composite index (SSEC) still has the lowest volatility spillover from other major stock markets (i.e., volatility spillover index of 3.76), but Korea Composite stock price index

(KOSPI) exhibits the lowest volatility spillover to other stock markets (i.e., volatility spillover index of 0.27).

In addition, Table 7 in the Appendix reports the magnitudes of volatility spillovers for realized semi-variance (RSV) and bi-power variation (BV) during the COVID-19 period respectively. The total spillover index increases to 83.64 and 87.70 for realized semi-variance and bi-power variation respectively, which is obviously larger than that of the full sample, indicating that the volatility spillover effect intensifies over the COVID-19 pandemic period, probably due to the rising risk uncertainty and risk contagion in the turbulent period (Su et al., 2023).

#### 4.4 Time–Frequency Volatility Spillovers for the Full Sample Period

Along with the overall connectedness among major stock market indices around the world, we also inspect the investment-horizon-specific connectedness among these markets. We follow the conventional standard and split the spectrum into three bands—short-term (i.e., one day to one week), medium-term (i.e., one week to one month), and long-term (i.e., over one month) (Kumar et al., 2022). Table 3 presents the results of volatility spillover index with realized variance (RV) in the short-term, medium-term, and long-term frequency bands, following Baruník and Krehlík (2018)'s frequency connectedness methodology.<sup>8</sup> The total spillover index in the short-, medium-, and long-term, as shown in the bottom right cell of Panel A to Panel C of the matrix, is 69.97, 82.22, and 87.52 respectively, indicating a fairly sizable level of volatility spillovers in the frequency domain and an increasing inter-connectedness with the larger frequency ranges.

Panel A to Panel C of Table 3 show the volatility connectedness and spillovers split between the frequency bands. In addition, ABS and WTN in Table 3 refer to absolute- and within- connectedness measures, respectively. Note that the pairwise spillover indices with realized variance for different frequency bands sum to the overall spillover index as shown in Table 2. The results suggest the dominance of the low frequency component (i.e., long-term horizon) in the spillover dynamics, which accounts for 33.97 (i.e., the sum of To ABS or From ABS in Panel C of Table 3) of the total spillover index of 80.55, i.e., almost 42.18% of the total spillover index. In contrast, the high frequency component (i.e., short-term horizon) in the spillover dynamics only accounts for 26.46% of the total spillover index (i.e., 21.31/80.55 = 26.46%). The untabulated results for BV and RSV provide similar evidence that the total spillovers among global stock markets are mostly driven by low frequency components for the full sample. Our results are consistent with the findings of Wang et al., (2022a, 2022b). Wang et al., (2022a, 2022b) show that the total spillovers among stock markets have been driven by low frequency components. However, the dominance of the long-term volatility connectedness

<sup>&</sup>lt;sup>8</sup> To measure volatility, we utilize realized variance (RV), realized semi-variance (RSV), and bi-power variation (BV), all calculated based on rolling sampling price series at 5-min intervals. However, due to space limitation, we omit the results for RSV and BV, which are available upon request by contacting the authors.

Table 3 Volat	ility spi	llover 1	measure	es for R	V—BK	K metho	q											
	AEX	BFX	FCHI	IXIC	SPX	N225	NSEI	RUT	I ITZ	FTSE	KS11	SSEC	OSEAX	GSPTSE	ISMS	FTMIB	FROM_ABS	FROM_WTH
Panel A: short	t time he	orizon-	—1–7 d	ays														
TO_ABS	2.18	1.29	1.98	1.79	0.15	0.98	0.41	1.37	2.71	1.95	0.40	0.09	0.95	2.49	0.90	1.68	21.31	N.A
TO_WTH	7.15	4.23	6.50	5.89	0.48	3.21	1.35	4.49	8.91	5.42	1.33	0.28	3.10	8.18	2.95	5.51	N.A	69.97
FROM_ABS	1.18	1.03	1.07	1.07	0.77	2.93	1.69	0.53	1.41	1.91	0.48	0.43	2.31	1.86	1.65	1.00	N.A	N.A
FROM_ WTH	3.87	3.37	3.50	3.51	2.54	9.63	5.55	1.75	4.62	6.26	1.57	1.40	7.57	6.12	5.43	3.28	N.A	N.A
Panel B: medi	um time	e horiz	on7	30 days														
TO_ABS	2.53	1.84	2.46	2.50	0.06	0.84	0.37	2.09	3.27	1.36	0.29	0.04	0.94	3.07	0.81	2.79	25.27	N.A
TO_WTH	8.24	6.00	8.02	8.13	0.19	2.73	1.19	6.80	10.63 4	4.43	0.96	0.15	3.05	9.98	2.65	9.09	N.A	82.22
FROM_ABS	1.90	1.93	1.87	1.92	1.25	0.88	1.46	1.89	1.76	1.53	1.77	1.00	1.14	1.42	1.64	1.90	N.A	N.A
FROM_ WTH	6.17	6.29	6.09	6.26	4.06	2.87	4.76	6.16	5.72	4.98	5.75	3.27	3.71	4.61	5.33	6.18	N.A	N.A
Panel C: long	time ho	nizon-	-beyon	d 30 da	ys													
TO_ABS	3.40	2.40	3.29	3.39	0.07	1.04	0.40	2.72	4.59	1.72	0.27	0.05	1.31	4.29	0.99	4.05	33.97	N.A
TO_WTH	8.76	6.17	8.49	8.74	0.17	2.67	1.02	7.02	11.82	4.4	0.68	0.12	3.39	11.05	2.55	10.44	N.A	87.52
FROM_ABS	2.45	2.56	2.56	2.27	3.03	0.70	2.58	2.95	2.11	1.86	3.39	0.59	1.16	1.90	1.64	2.24	N.A	N.A
FROM_ WTH	6.31	6.59	6.58	5.86	7.80	1.81	6.64	7.60	5.43	4.79	8.74	1.51	3.00	4.89	4.22	5.76	N.A	N.A
This table pre term, medium (RSV) and bi- WTN refer to	sents th 1-term, a -power absolute	e resul and lor variatic e and w	ts of vo ng-term on (BV, vithin, r	latility frequei ) are av espectiv	spillow ncy baı ailable vely. Th	ers in th nds, foll ; upon r he total	le freque lowing ] equest. spillove	ency do Baruník FROM r index	main ov c and Kı represe is highl	/er the v rehlík ( nts shoo ighted,	whole si 2018), 1 cks fror as show	ample p respecti n other /n in the	eriod. Pan vely. Due s, and TO bottom ri	el A to Pai to space li represents ght cell of	nel C sho mitation s shocks the mat	ow the vo , the resu to others ix of each	latility spillov llts for realize . The abbrevia h panel	ers in the short- 1 semi-variance ntions ABS and

in the stock market is in striking contrast to that of cryptocurrency market and commodity market, where most of the return and volatility connectedness occurs in the short-term (Ferrer et al., 2018; Kumar et al., 2022).

#### 4.5 Time–Frequency Volatility Spillovers During the COVID-19 Pandemic

Similarly, Table 4 presents the connectedness and spillover measures split between the frequency bands for the period between February 2020 and June 2022, i.e., the COVID-19 pandemic period. Foremost, the structure of the connectedness changes significantly when compared to the full sample period. Interestingly, we document the dominance of the volatility spillover over the medium- to long-term horizon as the low- to mediumfrequency components (i.e., approximately 7 to 30 days and over 30 days) in the spillover dynamics now accounts for as high as 78.04% of the total spillover index (i.e., 30.67/86.03=35.65% for the medium-term horizon and 36.46/86.03=42.37% for the long-term horizon respectively), which show that the volatility connectedness during this pandemic period is even more driven by medium-frequency components compared to that of the full sample period. That is, the magnification effects mostly apply to the medium-term frequency connectedness, i.e., the COVID-19 has a substantial impact on the connectedness among global stock markets in the medium- to long-term horizon. Similarly, cross-market spillovers of realized semi-variance (RSV) and bi-power variation (BV) magnify during the COVID-19 pandemic period and the frequency-based analysis suggests that the volatility spillover over the medium- to long-term horizon dominates that over the short-term horizon.9

Recent research has widely examined the impacts of the COVID-19 pandemic on volatility spillovers and unveiled that the COVID-19 outbreak induces finance contagion, probably due to the deteriorating economic situation and risk sentiment among global investors in the crisis period (Akhtaruzzaman et al., 2022; Corbet et al., 2022; Guo et al., 2021; Nguyen et al., 2021; Yarovaya et al., 2021a, 2021b). Unlike the conclusions drawn by Kumar et al. (2022) that the short-term horizon dominates in both returns and volatility spillovers in the cryptocurrency market during the COVID-19 period, we find that the COVID-19 pandemic has mostly strengthened the volatility connectedness among global stock markets over the medium- to long-term horizon, consistent with previous studies which have documented that crisis such as global financial crisis and Eurozone sovereign debt crisis (Mensi et al., 2018), and the outbreak of COVID-19 (Cheng et al., 2022) can lead to intensified volatility spillovers in global stock markets. Our results are in line with those of earlier studies. Wang et al., (2022a, 2022b) show that the total spillovers among stock markets driven by low frequency components peak at the end of March 2020 and then decline. However, our findings suggest that the COVID-19 has led to some fundamental changes in the global economy and its impact on contagion in financial markets can be long-lasting. Even with its importance regarding policy implications, there are few influential papers on this issue yet, which certainly needs to be addressed more intensively.

<sup>&</sup>lt;sup>9</sup> Due to space limitation, we omit the results for RSV and BV, which are available upon request by contacting the authors.

Table 4 Volat	ility sp	illover	measur	es for R	V duri	ng the C	-OUID-	19 outb	reak-I	BK meth	pou							
	AEX	BFX	FCHI	IXIC	SPX	N225	NSEI	RUT	STI	FTSE	KS11	SSEC	OSEAX	GSPTSE	SMSI	FTMIB	FROM_ABS	FROM_WTH
Panel A: shor	t time h	orizon	— <i>I</i> –7 a	lays														
TO_ABS	1.45	0.65	1.28	1.55	0.14	1.11	0.26	1.35	2.39	1.77	0.19	0.17	0.96	2.45	1.40	1.78	18.90	NA
TO_WTH	5.85	2.62	5.16	6.25	0.56	4.47	1.05	5.42	9.60	7.11	0.76	0.68	3.86	9.85	5.64	7.16	N.A	76.03
FROM_ABS	0.88	0.87	0.82	0.94	1.06	1.64	2.20	0.56	1.02	1.55	0.57	1.10	2.63	1.32	0.91	0.84	N.A	N.A
FROM_ WTH	3.54	3.49	3.28	3.77	4.27	6.61	8.86	2.26	4.12	6.22	2.29	4.43	10.57	5.32	3.65	3.36	N.A	N.A
Panel B: medi	ium tim	e horiz	-2uo	30 days														
TO_ABS	2.57	1.49	2.40	2.26	0.08	1.93	0.07	1.83	3.98	1.93	0.07	0.11	1.25	4.16	2.62	3.95	30.67	NA
TO_WTH	7.34	4.27	6.85	6.46	0.22	5.52	0.21	5.25	11.37	5.53	0.19	0.30	3.57	11.89	7.48	11.29	N.A	87.75
FROM_ABS	2.24	2.23	2.22	2.33	1.50	1.35	1.57	2.23	2.05	1.78	2.14	1.78	1.15	1.70	2.30	2.12	N.A	N.A
FROM_ WTH	6.40	6.37	6.35	6.66	4.28	3.85	4.48	6.38	5.88	5.09	6.12	5.08	3.30	4.88	6.58	6.06	N.A	N.A
Panel C: long	time h	orizon-	-beyon	d 30 da	уs													
TO_ABS	3.06	1.78	2.90	2.50	0.08	2.39	0.02	1.78	4.79	2.47	0.02	0.17	1.64	4.72	3.20	4.94	36.46	NA
TO_WTH	7.61	4.43	7.21	6.23	0.20	5.95	0.04	4.42	11.91	6.15	0.04	0.43	4.08	11.75	7.97	12.28	N.A	90.72
FROM_ABS	2.55	2.61	2.61	2.29	3.07	1.81	2.32	2.87	2.27	2.09	3.17	0.88	1.02	2.25	2.28	2.37	N.A	N.A
FROM_ WTH	6.35	6.49	6.50	5.71	7.63	4.51	5.77	7.13	5.65	5.20	7.90	2.20	2.53	5.59	5.67	5.90	N.A	N.A
This table pre the short-tern variance (RSV	sents th i, mediu /) and th	ne resul um-teri bi-powe	lts of vc m, and er variat	latility long-ter tion (BV	spillov m freg /) are <i>i</i>	ers in th luency h ivailable	he frequ bands, f	lency de ollowin equest	omain d ig Barur FROM	luring th ník and represe	le COV Krehlík nts shoc	ID-19 p; (2018), cks from	andemic p respectiv	eriod. Pan ely. Due to nd TO repr	el A to F space l' esents sh	anel C sh imitation, nocks to c	the results for the results for thers. The abb	ity spillovers in r realized semi- reviations ABS
	1 10 90	onnine (	מוזת אזת	teat (mm		1). 1110 (	ide mini			mgungı.		TIMOTIC		, 111 11 111 V		TO VINDIII	vavu pauvi	

Economic Policy Uncertainty and Volatility Spillovers Among...

#### 4.6 Robust Checks

In this subsection, we have conducted a series of robustness checks to check the sensitivity of our results. First, to ensure the same time lengths of estimation periods, we compare the results regarding the COVID-19 pandemic period with other periods of the same window length. Namely, we calculate Diebold and Yilmaz (2012)'s spillover index among the 16 global stock market indices for two sub-periods of the same time length, i.e., pre-COVID (February 2019 to January 2020) and COVID-19 period (February 2020 to January 2021). The results as shown in Table 8 in the Appendix suggest that the total volatility spillover index of the COVID-19 period is significantly larger than that of the pre-COVID-19 period (i.e., 88.61 versus 63.98), indicating a structural change in volatility spillovers among global stock markets since the outbreak of COVID-19. The untabulated results for realized semi-variance (RSV) and bi-power variation (BV) provide similar evidence that the volatility spillovers among global stock markets increase sharply during the turbulent period disrupted by the pandemic.

Second, we re-estimate the results by utilizing the rolling window method. However, the results are omitted for the sake of brevity. Again, the untabulated results confirm our conjecture that the volatility connectedness among global stock markets intensifies during turbulent periods, i.e., the COVID-19 period, as the estimated Diebold and Yilmaz (2012)'s spillover index is significantly larger over the COVID-19 pandemic period, compared to that of the full sample period.

Last but not the least, the number of parameters in the VAR model will increase dramatically as the number of markets under consideration increases, which may lead to the estimation infeasible when the sample size is small. For example, if the VAR model includes N = 17 markets and p = 2 lag periods, there will be 595 parameters to be estimated. Thus, traditional VAR models may experience insufficient degrees of freedom and encounter the so-called "curse of dimensionality" problem.

In order to address the above issue, following Demirer et al. (2018), we propose the least absolute shrinkage and selection operator (LASSO) to reduce the number of parameters to be estimated in the VAR model. The basic idea of LASSO method is to use regularization methodology to constrain the sum of absolute values, compress the values of some coefficients to 0 through penalty terms, and thereby obtain a model with much smaller degrees of freedom (Tibshirani, 1996). The code for implementing LASSO method using the R software is provided in Appendix 2. The results of the volatility spillover indices using LASSO method, as provided in Table 9 in the Appendix, show similar findings that the volatility spillovers among global stock markets increase sharply during the turbulent period of the COVID-19 pandemic.

### 4.7 EPU and Volatility Spillovers

The rapid spread of COVID-19 has caused unprecedented effects on global financial markets and substantially increased geopolitical risks and EPU worldwide. We further

investigate the effect of EPU on the connectedness between the 16 stock market indices examined in this study. For comparison, we examine the effect of the EPU during the entire period and during the COVID-19 pandemic. Namely, we conduct the OLS estimations over the full sample period and across different sub-periods.

To determine whether mean connectedness between stock markets varies through time and how it is affected by the COVID-19 pandemic, we estimate the time-varying connectedness measures using rolling-sample estimation, and perform OLS estimations separately. For the full sample period, the coefficient on EPU is positive and statistically significant. Similarly, for the COVID-19 pandemic period, we find a positive and statistically significant coefficient on EPU and Cases (Deaths) growth rate, as reported in Table 5, implying that volatility spillovers among major stock markets increase with an increase in global economic uncertainty as well as growth rate of confirmed COVID-19 cases (deaths). The empirical findings are generally in line with Erdem (2020), Engelhardt et al. (2021), and Cheng et al. (2022).

To examine the potentially confounding effect of EPU on stock market connectedness during the COVID-19 pandemic, we add the interaction terms between EPU and the growth rate of confirmed COVID-19 cases (deaths) per million inhabitants. We again find a positive and statistically significant coefficient on the interaction term between EPU and growth rates of confirmed COVID-19 cases (deaths), supporting our conjecture that volatility connectedness in global stock markets significantly increases when the growth rate of confirmed COVID-19 cases increases, which may be generated partly by increased selling pressure from panic-stricken investors.<sup>10</sup> Similar results are documented for the impact of the CBOE market volatility index (VIX) on connectedness before and during the COVID-19 outbreak.

# 5 Conclusion

We uncover the dynamic volatility spillovers in both time and frequency domains among 16 major stock market indices, spanning the period January 2015 to June 2022. Moreover, we examine the mediating effect of EPU on that relationship. We investigate those for the full sample period and the COVID-19 outbreak period (i.e., February 2020 to June 2022). The results show that there is substantial evidence of volatility spillovers in the global stock markets and that volatility connectedness is magnified during the turbulent period of COVID-19, which is in line with previous studies on various financial markets (Cheng et al., 2022; Kumar et al., 2022). However, the magnification effects mostly apply to the medium-term frequency connectedness, i.e., the COVID-19 has a strengthening impact on the connectedness among global stock markets mainly in the medium to long term, which have important implications for individual

<sup>&</sup>lt;sup>10</sup> Although not reported for reasons of brevity, we perform several robustness tests to ensure the validity of our results. First, we perform pooled OLS regressions for all stock markets examined in this study, where we include the two main effects as well as the interaction term, and find qualitatively similar results. Second, we use the connectedness index with realized semi-variance (RSV) and bi-power variation (BV) as our dependent variables and obtain similar results for both the full sample and the subsample of the COVID-19 period. Third, we utilize the overall index as well as the frequency-specific connectedness indices as the dependent variables and draw similar conclusions.

Dep. Var. =					
Total spillover index	(1)	(2)	(3)	(4)	(5)
Panel A: before the CO	VID-19 period				
Intercept	50.23***				
	(26.28)				
GEPU	0.039***				
	(5.12)				
Cases	N.A				
Deaths	N.A				
Cases * GEPU	N.A				
Deaths * GEPU	N.A				
VIX	0.035***				
	(19.08)				
Observations	389				
R-squared	0.450				
Panel B: during the CO	VID-19 period				
Intercept	65.93***	62.03***	61.93***	52.19***	58.88***
	(38.99)	(29.11)	(26.97)	(25.35)	(24.97)
GEPU	0.042***			0.033***	0.035***
	(7.22)			(4.99)	(5.76)
Cases		0.057***		0.028***	
		(8.99)		(5.32)	
Deaths			0.014**		0.000
			(2.02)		(1.34)
Cases * GEPU				0.035***	
				(5.88)	
Deaths * GEPU					0.022***
					(3.41)
VIX	0.022***	0.033***	0.034***	0.017***	0.012***
	(4.18)	(3.38)	(3.56)	(3.82)	(3.17)
Observations	73	73	73	73	73
R-squared	0.442	0.467	0.570	0.654	0.616

Table 5 The relationship between EPU and connectedness among global stock markets

This table presents the results regarding the relationship between EPU and volatility connectedness from OLS regressions over the pre-COVID-19 period in Panel A and the COVID-19 period (i.e., February 4, 2020 to June 30, 2022) in Panel B. The dependent variable is weekly rolling estimates of the total spillover index as calculated in Table 2. GEPU stands for the Global Economic Policy Uncertainty Index retrieved from FRED, Federal Reserve Bank of St. Louis. Cases (Deaths) is the log growth rate of confirmed COVID-19 cases (deaths) per million inhabitants. VIX is the stock market volatility index of CBOE. In columns (4) and (5) of Panel B, we additionally include the interaction between EPU and growth rates of confirmed COVID-19 cases (deaths). Standard errors corrected for heteroskedasticity and autocorrelation are reported in parentheses. \*, \*\* and \*\*\* indicates statistical significance at the 10%, 5% and 1% level respectively

investors, portfolio managers, policymakers, and central banks. Moreover, we find that EPU significantly affects the volatility spillovers among global stock market indices, particularly under the context of COVID-19.

Undoubtedly, a good understanding of volatility spillover among global financial markets can enhance the design of policies to contain it and improve the accountability of central banks, which allows policymakers to take appropriate actions to reduce the systemic risk and properly target the main factors responsible for driving those catastrophic events (Cappiello et al., 2015). However, this requires a clearer view of financial market interdependencies and volatility spillovers, as well as a deeper understanding of the incentives and motivations behind such relationships (Jackson & Pernoud, 2021). In the future, new analytical tools such as data mining, machine learning, as well as artificial intelligence could be applied to financial market interconnectedness and contagion analysis to form a holistic and systematic view of risk spillover and financial contagion.

There are several possible avenues for further research. First, it will be interesting to investigate the evolution of industry-level connectedness network for the full sample period and the COVID-19 outbreak period. For example, healthcare and tourism sectors might have vastly different responses to the shocks emanating from the COVID-19. Second, an intraday analysis of connectedness and volatility spillovers using tick-level data appears to be an interesting and promising solution, which allows for a more in-depth discussion on the mechanisms behind financial contagion. Last but not the least, the impact of economic policy uncertainty on risk contagion in financial markets should be considered more widely and comprehensively in academia and government practices.

# Appendix 1

See Appendix Tables 6, 7, 8, and 9.

Symbol	Description	Country (Region)
AEX	Amsterdam stock exchange index	Netherlands
BFX	Bell 20 index	Belgium
FCHI	CAC 40	France
FTMIB	FTSE MIB	Italy
FTSE	FTSE 100	UK
GSPTSE	S&P/TSX composite index	Canada
IXIC	Nasdaq 100	US
KS11	Korea composite stock price index (KOSPI)	Korea
N225	Nikkei 225	Japan
NSEI	NIFTY 50	India
OSEAX	Oslo exchange all-share index	Norway
RUT	Russel 2000	US
SMSI	Madrid general index	Spain
SPX	S&P 500 index	US
SSEC	Shanghai composite index	China
STI	Straits times index	Singapore

Table 6 List of stock market indices

This table lists the market indices adopted in this study

Table 7	Volatilit	ty spillov	er measu	res for <b>RS</b>	V and BV	-DY met	poq										
	AEX	BFX	FCHI	IXIC	SPX	N225	NSEI	RUT	STI	FTSE	KS11	SSEC	OSEAX	GSPTSE	SMSI	FTMIB	FROM
Panel A:	RSV														-	-	
Full sam	ple																
TO	7.78	5.06	7.15	4.30	0.37	2.39	0.64	6.13	9.72	5.77	1.38	0.29	3.72	9.63	2.04	10.50	76.88
FROM	5.47	5.21	5.39	5.45	5.03	4.11	5.78	5.15	5.23	4.91	5.24	1.82	4.26	4.98	4.04	4.79	N.A
NET	2.31	-0.15	1.76	-1.15	- 4.66	-1.72	-5.14	0.98	4.49	0.86	-3.86	-1.53	-0.54	4.65	-2.00	5.71	N.A
COVID-	19 perio	pq															
TO	69.9	3.37	5.80	4.16	0.31	2.90	0.25	4.92	10.35	5.69	0.79	0.18	3.86	10.89	11.00	12.49	83.64
FROM	5.66	5.45	5.60	5.73	5.65	4.31	6.07	5.57	5.36	5.14	5.71	3.29	4.60	5.20	5.21	5.12	N.A
NET	1.03	-2.08	0.20	-1.57	- 5.34	- 1.41	-5.82	-0.65	4.99	0.55	-4.92	-3.11	-0.74	5.69	5.79	7.37	N.A
Panel B:	BV																
Full sam	ple																
TO	7.90	6.31	7.82	8.24	0.27	3.73	0.69	6.55	69.6	4.99	1.15	0.04	4.38	7.41	6.48	6.65	82.28
FROM	5.58	5.58	5.55	5.29	4.84	4.97	5.76	5.48	5.32	5.46	5.72	1.21	5.12	5.52	5.48	5.40	N.A
NET	2.32	0.73	2.27	2.95	-4.57	-1.24	-5.07	1.07	4.37	-0.47	-4.57	- 1.17	-0.74	1.89	1.00	1.25	N.A
COVID-	19 perio	p															
TO	8.21	5.54	7.85	8.35	0.20	5.95	0.21	5.61	9.97	8.27	0.24	0.32	4.13	8.41	6.75	7.70	87.70
FROM	5.62	5.69	5.62	5.42	5.62	5.13	6.04	5.71	5.43	5.52	5.95	3.94	5.25	5.57	5.58	5.60	N.A
NET	2.59	-0.15	2.23	2.93	-5.42	0.82	- 5.83	-0.10	4.54	2.75	-5.71	- 3.62	-1.12	2.84	1.17	2.10	N.A
This tab. Yilmaz ( realized	le preser (2012)'s semi-va	its the res spillover riance (F	sults of v index in tSV) and	olatility spectrum of the time of the bi-power	oillovers au domain. T variation	nong glot he COVII (BV), all	al stock 1 D-19 peri calculate	narkets o od in this od based	ver the w study is on rollin	vhole san defined ig sampli	nple perioo as spannir ng price s	1 and dur 1g from F eries at :	ing the CO ebruary 4, 5-min inter	WID-19 peri 2020 to Jur vals, as pro	od, by uti le 30, 202 xies for v	lizing Diel 2. Here, w olatility m	bold and /e utilize neasures.
FRUM 1 direction	epresent	ts shocks ectedness	trom oth "TO" of	thers from	presents s "FROM"	hocks to ( If NET i	others, an s greater	d NET in than zero,	dicates t	he net dir ates that 1	ectional c the countr	onnected y is a net	ness of spi transmitte	llover, calcu r of shocks o	lated by s luring the	ubtracting period, of	the total herwise,
it will be	a net re	ceiver of	shocks.	The total s	spillover ir	idex is hig	hlighted,	as shown	in the b	ottom rig	ht cell of 1	he matriy	t of each pa	anel	)		

Table 8	Spillove	r measure	es for vo	latility b	efore and c	luring the	COVID-1	9 outbre	ak—RV								
	AEX	BFX	FCHI	IXIC	SPX	N225	NSEI	RUT	STI	FTSE	KS11	SSEC	OSEAX	GSPTSE	SMSI	FTMIB	FROM
Panel A:	pre – Cl	6I-GIAO	period														
TO	6.92	2.97	5.86	8.54	0.57	2.46	0.84	6.34	8.35	2.02	1.59	1.43	0.96	5.60	4.87	4.67	63.98
FROM	5.12	4.62	5.14	4.91	1.36	3.52	2.22	4.73	4.98	4.21	3.11	1.51	4.08	4.63	4.94	4.88	N.A
NET	1.80	-1.65	0.72	3.63	-0.79	-1.06	-1.38	1.61	3.37	-2.19	-1.52	-0.08	-3.12	0.97	-0.07	-0.21	N.A
Panel B:	COVID	-19 perio	$p_{i}$														
TO	7.44	5.79	6.62	6.01	0.60	5.36	0.39	5.34	10.38	7.15	0.30	1.09	5.18	10.76	6.38	9.81	88.61
FROM	5.69	5.71	5.72	5.72	5.90	5.08	6.13	5.78	5.48	5.42	6.06	4.58	4.86	5.41	5.62	5.47	N.A
NET	1.75	0.08	06.0	0.29	-5.30	0.28	- 5.74	-0.44	4.90	1.73	-5.76	- 3.49	0.32	5.35	0.76	4.34	N.A
This tab Yilmaz ( ning froi in total. price ser resents s "FROM" spillover	le presei 2012)'s n Febru: To meat ies at 5-i hocks to '. If NET index is	the respectively of the respective of the respective range of the result of the resul	esults of index in to Janus tility, we vals. To and NET er than z ted, as sh	volatilit t the time ary 2020 e utilize conserv indicate ero, it in town in t	y spillovei e domain. , and the ( realized v e space, w e space, w dicates th	rs among To ensure COVID-1 ariance (1 e omit thu lirectiona at the cou	global str e that the t 9 period i: RV), realin e results ff e results fa intry is a n of the ma	ock mark wo sub-s s defined zed semi- or RSV a dness of the transm trix of es	ets in the ample pe as spann variance nd BV, w spillover itter of s tch panel	e pre-CO' sriods are uing from (RSV), <i>i</i> vhich are vhich are , calculate thocks du	VID-19 al of the sar February und bi-pow available u available u subt subt ring the po	nd COVII ne time le 2020 to ver variat upon requ racting th racting th	D-19 perio ingth, the F January 20 ion (BV), est. FROM e total dire erwise, it v	d respective pre-COVID- 22, i.e., eac all calculate f represents ectional com vill be a net	ly, by uti 19 period i h period i d based o shocks fru nectedness receiver o	lizing Diel is defined ncludes 12 n rolling s om others, s "TO" oth of shocks."	oold and as span- as months ampling TO rep- ers from The total

	A UIAUILL	wonide (	u measu														
	AEX	BFX	FCHI	IXIC	SPX	N225	NSEI	RUT	ITZ	FTSE	KS11	SSEC	OSEAX	GSPTSE	ISMS	FTMIB	FROM
Panel A:	full sam	ple					-		-								
TO	78.87	71.54	78.43	70.20	7.43	40.36	25.07	66.69	106.50	88.98	13.24	7.60	61.53	71.72	75.07	82.99	949.52
FROM	82.47	80.03	82.33	78.01	21.83	57.87	55.92	80.14	81.98	78.86	36.66	24.03	71.80	80.93	80.14	78.51	N.A
NET	-3.60	- 8.49	-3.90	-7.81	-14.40	-17.51	- 30.85	-10.15	24.52	10.12	- 23.42	- 16.43	-10.27	-9.21	-5.07	4.48	N.A
Panel B.	· COVID-	-19 perio	p														
TO	106.12	93.10	104.77	89.03	11.11	53.90	26.52	79.60	105.06	90.11	40.54	10.61	75.20	88.28	81.23	71.41	1126.59
FROM	83.99	83.25	84.12	80.48	27.33	66.33	52.53	79.24	82.75	79.24	61.99	37.15	75.05	78.03	78.12	76.99	N.A
NET	22.13	9.85	20.66	8.56	-16.21	- 12.44	-26.01	0.36	22.31	10.87	- 21.45	-26.54	0.15	10.25	3.11	-5.59	N.A
This tab LASSO	le presen method.	Its the re The CO	sults of v VID-19 F	volatility beriod in	spillovers this study	among gle is defined	obal stock as spanni	markets of from I	during the February	e full sai 4, 2020	mple perio to June 30	od and the 0, 2022. T	cOVID- o measure	19 period r volatility,	espective we utilize	ly, by util e realized	izing the variance

(RV), realized semi-variance (RSV), and bi-power variation (BV), all calculated based on rolling sampling price series at 5-min intervals. To conserve space, we omit the results for RSV and BV, which are available upon request. FROM represents shocks from others, TO represents shocks to others, and NET indicates the net directional try is a net transmitter of shocks during the period, otherwise, it will be a net receiver of shocks. The total spillover index is highlighted, as shown in the bottom right cell connectedness of spillover, calculated by subtracting the total directional connectedness "TO" others from "FROM". If NET is greater than zero, it indicates that the coun-

of the matrix

ures for RV-I ASSO method 0 Table 9 Volatility snillover

# **Appendix 2: code for LASSO regression**

path = file.path(file.choose())
DATA = read.csv(path)
setwd (dir = "D:/R-4.3.0/workspace")

library(Spillover)

library(vars)

library(tidyverse)

library(xts)

library(devtools)

library(frequencyConnectedness)

library(ConnectednessApproach)

library(glmnet)

library(foreign)

DATA\$DATE <- as.Date(DATA\$DATE)

class(DATA\$DATE)

xtsdata0 = xts(DATA[,2:17], order.by = DATA[,"DATE"])

##Rolling estimation spilloverRollingDY12 LASSO

dca99 = ConnectednessApproach(xtsdata0,

nlag=4,

nfore=10,

window.size=250,

model="LASSO",

connectedness="Time",

 $Connectedness\_config=list(TimeConnectedness=list(generalized=TRUE)))$ 

table99 <- as.data.frame(dca99\$TABLE)

write.table(table99,"sample99.csv",sep=",")

PlotTCI(dca1) PlotTO(dca1) dca = ConnectednessTable(dca1)

write.table(table1,"sample1.csv",sep=",")

Acknowledgements Fei Su acknowledges financial support from the National Natural Science Fund of China (Grant No. 71901087) and the Fundamental Research Funds for the Central Universities (No. NR2022004). Yahua Xu acknowledges financial support under the Program for Innovation Research at Central University of Finance and Economics. All errors are our own.

#### Declarations

**Conflict of interest** The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# References

- Ait-Sahalia, Y., Mykland, P. A., & Zhang, L. (2005). How often to sample a continuous-time process in the presence of market microstructure noise. *The Review of Financial Studies.*, 18(2), 351–416. https://doi.org/10.1093/rfs/hhi016
- Akhtaruzzaman, M., Boubaker, S., Nguyen, D. K., & Rahman, M. R. (2022). Systemic risk-sharing framework of cryptocurrencies in the COVID–19 crisis. *Finance Research Letters*, 47, 102787. https://doi.org/10.1016/j.frl.2022.102787
- Albulescu, C. T., Demirer, R., Raheem, I. D., & Tiwari, A. K. (2019). Does the US economic policy uncertainty connect financial markets? Evidence from oil and commodity currencies. *Energy Economics*, 83, 375–388. https://doi.org/10.1016/j.eneco.2019.07.024
- Andersen, T. G. (2000). Some reflections on analysis of high-frequency data. Journal of Business and Economic Statistics, 18, 146–153. https://doi.org/10.2307/1392552
- Andersen, T. G., & Bollerslev, T. (1998). Deutsche mark–dollar volatility: Intraday activity patterns, macroeconomic announcements, and longer run dependencies. *The Journal of Finance*, 53(1), 219– 265. https://doi.org/10.1111/0022-1082.85732
- Andersen, T. G., Bollerslev, T., & Diebold, F. X. (2007). Roughing it up: Including jump components in the measurement, modeling, and forecasting of return volatility. *Review of Economics and Statistics.*, 89(4), 701–720. https://doi.org/10.1162/rest.89.4.701
- Andersen, T. G., Bollerslev, T., Diebold, F. X., & Labys, P. (2003). Modeling and forecasting realized volatility. *Econometrica*, 71(2), 579–625. https://doi.org/10.1111/1468-0262.00418
- Antonakakis, N., Chatziantoniou, I., & Filis, G. (2013). Dynamic co-movements of stock market returns, implied volatility and policy uncertainty. *Economics Letters*, 120(1), 87–92.
- Arouri, M., Estay, C., Rault, C., & Roubaud, D. (2016). Economic policy uncertainty and stock markets: Long-run evidence from the US. *Finance Research Letters*, 18, 136–141.
- Ashraf, B. N. (2020). Economic impact of government interventions during the COVID-19 pandemic: International evidence from financial markets. *Journal of Behavioral and Experimental Finance*, 27, 100371.
- Aslanidis, N., Bariviera, A. F., & Martinez-Ibanez, O. (2019). An analysis of cryptocurrencies conditional cross correlations. *Finance Research Letters*, 31, 130–137. https://doi.org/10.1016/j.frl.2019.04.019
- Badshah, I., Demirer, R., & Suleman, M. T. (2019). The effect of economic policy uncertainty on stockcommodity correlations and its implications on optimal hedging. *Energy Economics*, 84, 104553.
- Baker, S. R., Bloom, N., & Davis, S. J. (2022) Global Economic policy uncertainty index: Current price adjusted GDP [GEPUCURRENT], retrieved from FRED, Federal Reserve Bank of St. Louis, https://fred.stlouisfed.org/series/GEPUCURRENT
- Baker, S. R., Bloom, N., & Davis, S. J. (2016). Measuring economic policy uncertainty. *The Quarterly Journal of Economics*, 131(4), 1593–1636. https://doi.org/10.1093/qje/qjw024
- Balcilar, M., Roubaud, D., & Shahbaz, M. (2019). The impact of energy market uncertainty shocks on energy transition in Europe. *The Energy Journal*, 40(1), 55–79.
- Barndorff-Nielsen, O. E., Hansen, P. R., Lunde, A., & Shephard, N. (2008). Designing realized kernels to measure the ex-post variation of equity prices in the presence of noise. *Econometrica*, 76, 1481– 1536. https://doi.org/10.3982/ECTA6495

- Barndorff-Nielsen, O. E., Kinnebrock, S., & Shephard, N. (2010). Measuring downside risk ealized semivariance. In T. Bollerslev, J. Russell, & M. Watson (Eds.), *Volatility and time series* econometrics: Essays in honor of Robert F. Engle. New York: Oxford University Press.
- Barndorff-Nielsen, O. E., & Shephard, N. (2004). Power and bipower variation with stochastic volatility and jumps. *Journal of Financial Econometrics*, 2, 1–48. https://doi.org/10.1093/jjfinec/nbh001
- Barndorff-Nielsen, O. E., & Shephard, N. (2006). Econometrics of testing for jumps in financial economics using bipower variation. *Journal of Financial Econometrics*, 4, 1–30. https://doi.org/10. 1093/jjfinec/nbi022
- Baruník, J., & Křehlík, T. (2018). Measuring the frequency dynamics of financial connectedness and systemic risk. *Journal of Financial Econometrics*, 16(2), 271–296.
- Baruník, J., Kočenda, E., & Vácha, L. (2016). Symmetric connectedness on the U.S. stock market: Bad and good volatility spillovers. *Journal of Financial Markets*, 27, 55–78.
- Baruník, J., Kočenda, E., & Vácha, L. (2017). Asymmetric volatility connectedness on the forex market. Journal of International Money and Finance, 77, 39–56.
- Belke, A., Dubova, I., & Osowski, T. (2016). Policy uncertainty and international financial markets: the case of Brexit. *Applied Economics*, 50(34–35), 3752–3770.
- BenSaida, A., Litimi, H., & Abdallah, O. (2018). Volatility spillover shifts in global financial markets. *Economic Modelling*, 73, 343–353.
- Bollerslev, T., Hood, B., Huss, J., & Pedersen, L. H. (2018). Risk everywhere: Modeling and managing volatility. *Review of Financial Studies.*, 31, 2729–2773. https://doi.org/10.1093/rfs/hhy041
- Bouri, E., Lucey, B., & Roubaud, D. (2020). The volatility surprise of leading cryptocurrencies: Transitory and permanent linkages. *Finance Research Letters*, 33, 101188. https://doi.org/10.1016/j. frl.2019.05.006
- Cappiello, L., Fache Rousová, L., & Montagna, M. (2015). Systemic risk, contagion and financial networks. *Financial Stability Review*, 2, 146–157.
- Cheng, T., Liu, J., Yao, W., & Zhao, A. B. (2022). The impact of COVID-19 pandemic on the volatility connectedness network of global stock market. *Pacific-Basin Finance Journal*, 71, 101678. https:// doi.org/10.1016/j.pacfin.2021.101678
- Christou, C., Cunado, J., Gupta, R., & Hassapis, C. (2017). Economic policy uncertainty and stock market returns in PacificRim countries: Evidence based on a Bayesian panel VAR model. *Journal of Multinational Financial Management*, 40, 92–102.
- Ciaian, P., Rajcaniova, M., & Kancs, D. (2018). Virtual relationships: Short- and long-run evidence from Bitcoin and altcoin markets. *Journal of International Financial Markets, Institutions and Money*, 52, 173–195. https://doi.org/10.1016/j.intfin.2017.11.001
- Corbet, S., Hou, Y., Hu, Y., & Oxley, L. (2022). The influence of the COVID-19 pandemic on the hedging functionality of Chinese financial markets. *Research in International Business and Finance*, 59, 101510. https://doi.org/10.1016/j.ribaf.2021.101510
- Dai, P., Xiong, X., Liu, Z., Huynh, T. D., & Sun, J. (2021). Preventing crash in stock market: The role of economic policy uncertainty during COVID-19. *Financial Innovation*, 7(1), 1–15.
- Das, N. M., Rout, B. S., & Khatun, Y. (2023). Does G7 engross the shock of COVID 19: An assessment with market volatility? Asia-Pacific Financial Markets, 30, 795–816.
- Diebold, F. X., & Yilmaz, K. (2009). Measuring financial asset return and volatility spillovers, with application to global equity markets. *Economic Journal*, 119(534), 158–171. https://doi.org/10. 1111/j.1468-0297.2008.02208.x
- Diebold, F. X., & Yilmaz, K. (2012). Better to give than to receive: Predictive directional measurement of volatility spillovers. *International Journal of Forecasting*, 28(1), 57–66. https://doi.org/10.1016/j. ijforecast.2011.02.006
- Diebold, F. X., & Yilmaz, K. (2014). On the network topology of variance decompositions: Measuring the connectedness of financial firms. *Journal of Econometrics*, 182(1), 119–134. https://doi.org/10. 1016/j.jeconom.2014.04.012
- Engelhardt, N., Krause, M., Neukirchen, D., & Posch, P. N. (2021). Trust and stock market volatility during the COVID-19 crisis. *Finance Research Letters*, 38, 101873. https://doi.org/10.1016/j.frl. 2020.101873
- Erdem, O. (2020). Freedom and stock market performance during Covid-19 outbreak. *Finance Research Letters*. https://doi.org/10.1016/j.frl.2020.101671
- Fang, L., Yu, H., & Li, L. (2017). The effect of economic policy uncertainty on the long-term correlation between U.S. stock and bond markets. *Economic Modelling*, 66(C), 139–145.

- Fang, L., Bouri, E., Gupta, R., & Roubaud, D. (2018). Does global economic uncertainty matter for the volatility and hedging effectiveness of Bitcoin? *International Review of Financial Analysis*, 61, 39–36.
- Ferrer, R., Shahzad, S. J. H., López, R., & Jareño, F. (2018). Time and frequency dynamics of connectedness between renewable energy stocks and crude oil prices. *Energy Economics*, 76, 1–20. https://doi.org/10.1016/j.eneco.2018.09.022
- Gamba-Santamaria, S., Gomez-Gonzalez, J. E., Hurtado-Guarin, J. L., & Melo-Velandia, L. F. (2019). Volatility spillovers among global stock markets: Measuring total and directional effects. *Empirical Economics*, 56(5), 1581–1599.
- Greenwood-Nimmo, M., Nguyen, V. H., & Rafferty, B. (2016). Risk and return spillovers among the G10 currencies. *Journal of Financial Markets*, 31, 43–62.
- Guo, X., Lu, F., & Wei, Y. (2021). Capture the contagion network of bitcoin—Evidence from pre and mid COVID-19. Research in International Business and Finance, 58, 101484. https://doi.org/10.1016/j. ribaf.2021.101484
- He, F., Wang, Z., & Yin, L. (2020). Asymmetric volatility spillovers between international economic policy uncertainty and the U.S. stock market. *The North American Journal of Economics and Finance*, 51, 2101084.
- Hu, Z., Kutan, A. M., & Sun, P. W. (2018). Is US economic policy uncertainty priced in China's A-shares market? Evidence from market, industry, and individual stocks. *International Review of Financial Analysis*, 57, 207–220. https://doi.org/10.1016/j.irfa.2018.03.015
- Iyer, S. R., & Simkins, B. J. (2022). COVID-19 and the Economy: Summary of research and future directions. *Finance Research Letters*, 47(B), 102801. https://doi.org/10.1016/j.frl.2022.102801
- Jackson, M. O., & Pernoud, A. (2021). Systemic risk in financial networks: A survey. Annual Review of Economics, 13, 171–202. https://doi.org/10.1146/annurev-economics-083120-111540
- Ji, H., Wang, H., Xu, J., & Liseo, B. (2020). Dependence structure between China's stock market and other major stock markets before and after the 2008 financial crisis. *Emerging Markets Finance and Trade*, 56(11), 2608–2624. https://doi.org/10.1080/1540496X.2019.1615434
- Ji, Q., Bouri, E., Lau, C. K. M., & Roubaud, D. (2019b). Dynamic connectedness and integration in cryptocurrency markets. *International Review of Financial Analysis*, 63, 257–272.
- Ji, Q., Bouri, E., Roubaud, D., & Kristoufek, L. (2019a). Information interdependence among energy, cryptocurrency and major commodity markets. *Energy Economics*, 81, 1042–1055.
- Karanasos, M., & Yfanti, S. (2020). On the macro-drivers of realized volatility: The destabilizing impact of UK policy uncertainty across Europe. *European Journal of Finance*, 26(12), 1146–1183.
- Katsiampa, P., Corbet, S., & Lucey, B. (2019). Volatility spillover effects in leading cryptocurrencies: A BEKK-MGARCH analysis. *Finance Research Letters*, 29, 68–74. https://doi.org/10.1016/j.frl.2019. 03.009
- Kitano, S., & Zhou, Y. (2022). Effects of China's capital controls on individual asset categories. Finance Research Letters, 49, 103032.
- Kumar, A., Iqbal, N., Mitra, S. K., Kristoufek, L., & Bouri, E. (2022). Connectedness among major cryptocurrencies in standard times and during the COVID-19 outbreak. *Journal of International Financial Markets, Institutions & Money*, 77, 101523.
- Li, X. M., & Peng, L. (2017). US economic policy uncertainty and co-movements between Chinese and US stock markets. *Economic Modelling*, 61, 27–39.
- Li, X. L., Balcilar, M., Gupta, R., & Chang, T. (2015). The causal relationship between economic policy uncertainty and stock returns in China and India: Evidence from a bootstrap rolling window approach. *Emerging Markets Finance and Trade*, 52(3), 674–689. https://doi.org/10.1080/15404 96X.2014.998564
- Li, R., Li, S., Yuan, D., Chen, H., & Xiang, S. (2023). Spillover effect of economic policy uncertainty on the stock market in the post-epidemic era. *The North American Journal of Economics and Finance*, 64, 101846.
- Li, X., Shen, D., & Zhang, W. (2018). Do Chinese internet stock message boards convey firm-specific information? *Pacific-Basin Finance Journal*, 49, 1–14.
- Li, Y., Luo, J., & Jiang, Y. (2021). Policy uncertainty spillovers and financial risk contagion in the Asia-Pacific network. *Pacific-Basin Finance Journal*, 67, 101554.
- Li, Y., Shi, Y., Shi, Y., Yi, S., & Zhang, W. (2023). COVID-19 vaccinations and risk spillovers: Evidence from Asia-Pacific stock markets. *Pacific-Basin Finance Journal*, 79, 102004.

- Lin, S., & Chen, S. (2021). Dynamic connectedness of major financial markets in China and America. International Review of Economics and Finance, 75, 646–656. https://doi.org/10.1016/j.iref.2021. 04.033
- Liu, L. Y., Patton, A. J., & Sheppard, K. (2015). Does anything beat 5-minute RV? A comparison of realized measures across multiple asset classes. *Journal of Econometrics.*, 187(1), 293–311. https:// doi.org/10.1016/j.jeconom.2015.02.008
- Liu, Y., Wei, Y., Wang, Q., & Liu, Y. (2021). International stock market risk contagion during the COVID-19 pandemic. *Finance Research Letters*, 45, 102145.
- Ma, R., Liu, Z., & Zhai, P. (2022). Does economic policy uncertainty drive volatility spillovers in electricity markets: Time and frequency evidence. *Energy Economics*, 107, 105848.
- Matkovskyy, R., Akanksha, J., & Michael, D. (2020). Effects of economic policy uncertainty shocks on the interdependence between Bitcoin and traditional financial markets. *The Quarterly Review of Economics and Finance*, 77(C), 150–155.
- McAleer, M., & Medeiros, M. C. (2008). Realized volatility: A review. *Econometric Reviews.*, 27(1–3), 10–45. https://doi.org/10.1080/07474930701853509
- Mclver, R. P., & Kang, S. H. (2020). Financial crises and the dynamics of the spillovers between the U.S. and BRICS stock markets. *Research in International Business and Finance*, 54, 101276.
- Mei, D., Zeng, Q., & Zhang, Y. (2018). Does US economic policy uncertainty matters for European stock markets volatility? *Physica A: Statistical Mechanics and its Applications*, 512, 215–221.
- Mensi, W., Boubaker, F. Z., Al-Yahyaee, K. H., & Kang, S. H. (2018). Dynamic volatility spillovers and connectedness between global, regional, and GIPSI stock markets. *Finance Research Letters*, 25, 230–238. https://doi.org/10.1016/j.frl.2017.10.032
- Moratis, G. (2021). Quantifying the spillover effect in the cryptocurrency market. *Finance Research Letters*, 38, 101534. https://doi.org/10.1016/j.frl.2020.101534
- Nguyen, D. T., Phan, D. H. B., Ming, T. C., & Nguyen, V. L. (2021). An assessment of how COVID-19 changed the global equity market. *Economic Analysis and Policy*, 69, 480–491. https://doi.org/10. 1016/j.eap.2021.01.003
- Phan, D. H. B., Sharma, S. S., & Tran, V. T. (2018). Can economic policy uncertainty predict stock returns? Global evidence. *Journal of International Financial Markets, Institutions and Money*, 55, 134–150.
- Qian, L., Jiang, Y., Long, H., & Song, R. (2020). The roles of Economic policy uncertainty and the COVID-19 pandemic in the correlation between cryptocurrency and stock markets. *The Singapore Economic Review*, 4, 1–30. https://doi.org/10.1142/S0217590820470062
- Rehan, M., Alvi, J., & Karaca, S. S. (2022). Short term stress of Covid-19 on world major stock indices. Asia-Pacific Financial Markets, 29, 527–568.
- Sharif, A., Aloui, C., & Yarovaya, L. (2020). COVID-19 pandemic, oil prices, stock market, and policy uncertainty nexus in the US economy: Fresh evidence from the wavelet-based approach. *International Review of Financial Analysis*, 70, 101496. https://doi.org/10.1016/j.irfa.2020.101496
- Shen, D., & Zhang, W. (2021). Stay-at-home stocks versus go-outside stocks: The impacts of COVID-19 on the Chinese stock market. Asia-Pacific Financial Markets, 28(2), 305–318.
- Shi, Y., & Wang, L. (2023). Comparing the impact of Chinese and U.S. economic policy uncertainty on the volatility of major global stock markets. *Global Finance Journal*, 57, 100860.
- So, M. K. P., Chu, A. M. Y., & Chan, T. W. C. (2021). Impacts of the COVID-19 pandemic on financial market connectedness. *Finance Research Letters*, 38, 101864. https://doi.org/10.1016/j.frl.2020. 101864
- Su, F., & Wang, X. (2021). Investor co-attention and stock return co-movement: Evidence from China's A-share stock market. North American Journal of Economics and Finance, 58, 101548.
- Su, F., Zhai, L., Zhou, Y., Zhuang, Z., & Wang, F. (2023). Risk contagion in financial markets: A systematic review using bibliometric methods. *Australian Economic Papers*. https://doi.org/10. 1111/1467-8454.12301
- Tibshirani, R. (1996). Regression shrinkage and selection via the LASSO. *Journal of the Royal Statistical Society: Series B (methodological)*, 58(1), 267–288.
- Tiwari, A. K., Aye, C., Gupta, R., & Gkillas, K. (2020). Gold-oil dependence dynamics and the role of geopolitical risks: evidence from a Markov-switching time-varying copula model. *Energy Economics*, 88, 104748.
- Wang, J., Lu, X., He, F., & Ma, F. (2020). Which popular predictor is more useful to forecast international stock markets during the coronavirus pandemic: VIX vs EPU? *International Review of Financial Analysis*, 72, 101596.

- Wang, C., Xiong, X., & Shen, D. (2022a). Tail risks, firm characteristics, and stock returns. *Pacific-Basin Finance Journal*, 75, 101854.
- Wang, D., Li, P., & Huang, L. (2022b). Time-frequency volatility spillovers between major international financial markets during the COVID-19 pandemic. *Finance Research Letters*, 46(A), 102244.
- Wang, G.-J., Xie, C., Wen, D., & Zhao, L. (2019). When Bitcoin meets economic policy uncertainty (EPU): Measuring risk spillover effect from EPU to Bitcoin. *Finance Research Letters*, 31, 489– 497. https://doi.org/10.1016/j.frl.2018.12.028
- Xiong, X., Bian, Y., & Shen, D. (2018). The time-varying correlation between policy uncertainty and stock returns: Evidence from China. *Physica A: Statistical Mechanics and its Applications*, 499, 413–419.
- Yadav, M. P., Sharma, S., & Bhardwaj, I. (2023). Volatility spillover between chinese stock market and selected emerging economies: A dynamic conditional correlation and portfolio optimization perspective. *Asia-Pacific Financial Markets*, 30, 427–444. https://doi.org/10.1007/ s10690-022-09381-9
- Yarovaya, L., Brzeszczynski, J., Goodell, J. W., Lucey, B., & Lau, C. K. (2021a). Rethinking financial contagion: Information transmission mechanism during the COVID-19 pandemic. Available at SSRN: http://ssrn.com/abstract=3602973
- Yarovaya, L., Matkovskyy, R., & Akanksha, J. (2021b). The effects of a "black swan" event (COVID-19) on herding behavior in cryptocurrency markets. *Journal of International Financial Markets, Institutions and Money*, 75, 101321. https://doi.org/10.1016/j.intfin.2021.101321
- Yarovaya, L., Matkovskyy, R., & Jalan, A. (2022). The COVID-19 black swan crisis: Reaction and recovery of various financial markets. *Research in International Business and Finance*, 59, 101521. https://doi.org/10.1016/j.ribaf.2021.101521
- Yi, Y., Ma, F., Zhang, Y., & Huang, D. (2018). Forecasting the prices of crude oil using the predictor, economic and combined constraints. *Economic Modelling*, 75, 237–245.
- Youssef, M., Mokni, K., & Ajmi, A. N. (2021). Dynamic connectedness between stock markets in the presence of the COVID-19 pandemic: Does economic policy uncertainty matter? *Financial Innovation*. https://doi.org/10.1186/s40854-021-00227-3
- Yu., M., & Song, J. (2018). Volatility forecasting: Global economic policy uncertainty and regime switching. *Physica A: Statistical Mechanics and its Applications*, 511(C), 316–323.
- Yu, H., Lv, W., Liu, H., & Wang, J. (2022). Economic policy uncertainty and corporate bank credits: Evidence from China. *Emerging Markets Finance and Trade*, 58(11), 3023–3033. https://doi.org/10. 1080/1540496X.2021.2021179
- Zhang, D., Hu, M., & Ji, Q. (2020a). Financial markets under the global pandemic of COVID-19. Finance Research Letters. https://doi.org/10.1016/j.frl.2020.101528
- Zhang, Y., Chen, R., & Ma, D. (2020b). A weighted and directed perspective of global stock market connectedness: A variance decomposition and GERGM framework. *Sustainability*, 12(11), 4605. https://doi.org/10.3390/su12114605
- Zhang, Y., & Yan, X. (2020). The impact of US economic policy uncertainty on WTI crude oil returns in different time and frequency domains. *International Review of Economics and Finance*, 69, 750– 768. https://doi.org/10.1016/j.iref.2020.04.001

**Publisher's Note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Springer Nature or its licensor (e.g. a society or other partner) holds exclusive rights to this article under a publishing agreement with the author(s) or other rightsholder(s); author self-archiving of the accepted manuscript version of this article is solely governed by the terms of such publishing agreement and applicable law.